Author's response to reviews

Title: Reliability of upright posture measurements in primary school children

Authors:

Maureen P McEvoy (maureen.mcevoy@unisa.edu.au)
Karen Grimmer (karen.grimmer@unisa.edu.au)

Version: 7 Date: 24 June 2005

Author's response to reviews:

The authors have addressed all formatting changes requested. It is hoped that these changes meet with the requirements of the Editorial Committee.